Author's response to reviews

Title: The effects of a home-based arm ergometry exercise programme on physical fitness, fatigue and activity in Polio survivors: protocol for a randomised controlled trial

Authors:

Deirdre E Murray (deirdremurray@beaumont.ie)
Dara Meldrum (dmeldrum@rcsi.ie)
Roisin Moloney (roisinmoloney@beaumont.ie)
Frances Horgan (fhorgan@rcsi.ie)
Orla Hardiman (orla@hardiman.net)

Version: 2 Date: 26 October 2012

Author’s response to reviews:

Dear Editor,

Please find uploaded the revised version of the manuscript which was reviewed by you via email. If possible, I would also like to add an author, Anna Campion, as she has been recruited as an additional blinded assessor who is making significant contributions to the ongoing research.

The study has enrolled 42 participants to date, with 5 further volunteers recruited.

Thank you for your assistance to date,

Deirdre Murray